Author’s response to reviews

Title: Study Protocol: ASCRIBED: The impact of Acute SystematiC inflammation upon cerebrospinal fluid and blood Biomarkers of brain inflammation and injury in Dementia: a study in acute hip fracture patients

Authors:

Nick Leavey (nickleavey@hotmail.com)
Simon Hammond (S.Hammond@uea.ac.uk)
Lee Shepstone (L.Shepstone@uea.ac.uk)
Jane Cross (J.Cross@uea.ac.uk)
Henrik Zetterberg (henrik.zetterberg@clinchem.gu.se)
Colm Cunningham (Cunninco@tcd.ie)
Alasdair MacLullich (A.Maclullich@ed.ac.uk)
Leiv Otto (l.o.watne@gmail.com)
Anne Marie Minihane (A.Minihane@uea.ac.uk)
Clive Ballard (Clive.Ballard@kcl.ac.uk)
Anne-Brita Knapskog (anne-brita@knapskog.net)
Roanna Hall (roanna.hall@nhslothian.scot.nhs.uk)
Gregory Howard (Gregory.Howard@uea.ac.uk)
Matt Hammond (M.Hammond@uea.ac.uk)
Chris Fox (Chris.Fox@uea.ac.uk)

Version: 2 Date: 22 Aug 2019

Author’s response to reviews:

Tables and figure reloaded